DOCUMENT DE TREBALL

XREAP2011-09

The link between public support and private R&D effort: What is the optimal subsidy?

Nestor Duch-Brown (IEB) José García-Quevedo (IEB) Daniel Montolio (IEB)



The link between public support and private R&D effort: What is the optimal subsidy?

Néstor Duch-Brown

Department of Public Economy, Political Economy and Spanish Economy and Barcelona Institute of Economics (IEB), University of Barcelona. Av Diagonal 690. 08034 Barcelona (Spain) E-mail: nduch@ub.edu

José García-Quevedo

Department of Public Economy, Political Economy and Spanish Economy and Barcelona Institute of Economics (IEB), University of Barcelona. Av Diagonal 690. 08034 Barcelona (Spain) E-mail: jgarciag@ub.edu

L-man. Jgarenaq@u0.edt

Daniel Montolio

Department of Public Economy, Political Economy and Spanish Economy and Barcelona Institute of Economics (IEB), University of Barcelona. Av Diagonal 690. 08034 Barcelona (Spain) E-mail: montolio@ub.edu

Abstract The effectiveness of R&D subsidies can vary substantially depending on their characteristics. Specifically, the amount and intensity of such subsidies are crucial issues in the design of public schemes supporting private R&D. Public agencies determine the intensities of R&D subsidies for firms in line with their eligibility criteria, although assessing the effects of R&D projects accurately is far from straightforward. The main aim of this paper is to examine whether there is an optimal intensity for R&D subsidies through an analysis of their impact on private R&D effort. We examine the decisions of a public agency to grant subsidies taking into account not only the characteristics of the firms but also, as few previous studies have done to date, those of the R&D projects. In determining the optimal subsidy we use both parametric and non-parametric techniques. The results show a non-linear relationship between the percentage of subsidy received and the firms' R&D effort. These results have implications for technology policy, particularly for the design of R&D subsidies that ensure enhanced effectiveness.

JEL: O38, H32 Keywords: R&D, public subsidies, evaluation

Acknowledgements We thank ACC10 (Government of Catalonia), and in particular its research department for the information used in this paper and their help in data treatment and management. José García-Quevedo acknowledges support from grants ECO2010-16934 and 2009SGR102. Daniel Montolio acknowledges support form grant 2009SGR102. A previous version of this paper was presented at the 12th European Network on Industrial Policy (EUNIP) International Conference (Reus, Spain, June, 2010) and at the Zvi Griliches Research Seminar (Barcelona Graduate School of Economics, Spain, July, 2010). We are grateful for the comments of participants, particularly Isabel Busom, Elena Huergo and Carlos J. Serrano. The usual disclaimer applies.

1. Introduction

Support for R&D and innovation plays a central role in technology policies and is based on arguments for which a wide consensus exists. Among the justifications for public intervention the most fundamental is market failure, which can result in R&D investment falling below what is socially optimal in the absence of public intervention (Arrow, 1962). Governments, therefore, take different actions to support private technological activities, with R&D subsidies being one of the most frequently adopted tools.

Although the reasons for public support to private R&D are well established, its effectiveness needs to be examined. In recent years, public policy evaluation has acquired growing importance and although the results are not entirely conclusive, recent literature reviews (David *et al.*, 2000; García-Quevedo, 2004) indicate that, in the case of R&D subsidies, the existence of an additional effect is the most frequent outcome.

Most studies evaluating R&D subsidies tend to analyze the average effects of these subsidies on recipient firms. Although a number of these papers take into account certain firm characteristics in determining whether the effects are homogenous, very few analyse whether these effects vary depending on the specific characteristics of the R&D subsidies themselves. As Takalo *et al.* (2005) point out, surprisingly, very little is known about the way in which R&D subsidies are allocated by public agencies. The degree of effectiveness of R&D policy can differ substantially depending on the allocation procedure and on the characteristics of R&D subsidies. Among these characteristics, the amount and intensity of the public subsidy (defined as the gross subsidy amount expressed as a percentage of the project's eligible costs) are relevant aspects in the design of public schemes in support of private R&D.

In this respect, firms' R&D projects vary considerably and so the optimal public R&D policy will vary from project to project. This means that, although the rules for allocating R&D subsidies are the same for all firms, the policy should be heterogeneous *ex-post*, i.e., granting different subsidy intensities according to the characteristics of the projects and their expected effects (Toivanen, 2006). Thus, public agencies determine different intensities of R&D subsidies to firms, taking into account EU rules and

depending on established eligibility criteria. The European Union, in the Community framework for State aid for Research and Development and Innovation, emphasizes the importance of carefully defining state aid measures so as to avoid distortion and to guarantee their effectiveness. Specifically, it stresses that the amount and intensity of the subsidy should be limited to the minimum required for the R&D activity to take place. Such an allocation faces the difficulty of determining the optimal subsidy intensity, given the complexity of assessing accurately the effects and impact of R&D projects.

As suggested by a number of recent empirical studies (González *et al.*, 2005; Görg and Strobl, 2007; Aschhoff, 2009; Cerulli and Poti, 2010), the effects of subsidies can differ substantially depending on their characteristics. Although these contributions to the literature provide valuable information, our knowledge about the effects that the distribution of the subsidies, in terms of intensity, have on the behaviour of recipient firms and, eventually, on the impact and efficiency of these public programmes is scarce.

The objective of this paper, therefore, is to contribute to this literature by examining whether there is an optimal intensity for public R&D subsidies. In order to do this, we analyse the degree of financial additionality in terms of private R&D effort for the different intensities of subsidy granted to firms' R&D projects. Moreover, we analyse the public agency's decisions to grant the subsidy taking into account not only the characteristics of the firms but also those of the R&D projects themselves. As a few recent papers (Takalo *et al.*, 2008; Huergo and Trenado, 2010) show, the characteristics of the firms. The availability of a detailed database with information on the characteristics of private R&D projects (including the scores given by the public agency to each R&D project), the applicant firms, and the subsidies granted allows us to estimate the optimal subsidy that firms should receive from the public agency in order to obtain a higher R&D additionality.

For this purpose the empirical strategy proposed here consists in estimating parametrically and non-parametrically the impact of R&D subsidy intensity as a determinant of the firms' R&D effort. In the parametric set-up, and given the possible

endogeneity problems of R&D subsidy intensity we make use of a two-step procedure with a Tobit model with instrumental variables in the first stage (see Wallsten, 2000), and a Heckman correction model in the second stage. In the non-parametric set-up, the use of a conditional quantile regression approach allows us to estimate the effect of subsidy intensity for different types of recipient firm. Sorting firms by R&D effort, we obtain the impact of public subsidy intensity conditional on the R&D effort distribution; therefore, we observe the possible existence of an optimal R&D subsidy for each category of firm.

The rest of the paper is organised as follows. Section 2 reviews both the theoretical and the empirical literature and identifies the main hypotheses to be tested. Section 3 deals with the empirical strategy. Section 4 presents in detail the database used. Section 5 is devoted to present and discuss the main results obtained. Finally, section 6 presents the conclusions with some implications for the design of innovation policy.

2. Literature review: Theoretical hypotheses and empirical evidence

A relevant issue on the current empirical research agenda in the economics of technical change is the estimation of the "additionality effect", the change in privately financed R&D expenditures induced by government subsidization of R&D activities. However, to date, there have been few theoretical contributions formulating models of firm decision-making that take into account the mechanism by which this effect can occur (David et al., 2000; Takalo et al., 2008; Lee, 2011).

The arguments justifying why subsidies can stimulate private R&D expenditures are well known. First, upon receiving the agency's approval for its project, the firm commits to match, in a given proportion, the subsidy received, thus inducing subsidized firms, ceteris paribus, to increase their own R&D expenditure. Second, a subsidy to a particular project can lower the fixed costs and, hence, increase the profitability of other current (and future) R&D projects. Finally, learning and know-how spillovers from the subsidized project to other current (and future) R&D projects can increase the profitability.

What if the firm would have undertaken a particular subsidized R&D project even if the subsidy had not been granted? In this case, from the public agency's point of view the subsidy is superfluous since it would be fully crowding out private R&D expenditures. Consequently, an increase in privately financed R&D can only be guaranteed if the projects approved are those that would not have been undertaken without the subsidy. This vital information is, of course, usually unknown to the agency and constitutes a potential source of inefficiency.

This result is heavily dependent on the strong and unrealistic assumption that the firm will not change its R&D projects' portfolio after receiving a subsidy. For instance, even if the subsidy is superfluous, the private funds released by the subsidy could be used to finance other R&D projects, thus leaving its own R&D expenditures unaffected. Besides, receiving a subsidy may serve as a signal to lower the cost of new funds. In this case, the firm may accelerate existing projects or start new ones, thereby, increasing privately financed R&D expenditures. This could occur even if the subsidy to a particular project is superfluous.

Moreover, even in instances when the subsidized project would not have been undertaken without a subsidy, privately financed R&D expenditures may fall. In cases where a firm lacks enough skilled R&D workers or faces certain liquidity constraints, it may be too costly to implement the non-subsidized projects along with the subsidized project to which it is committed. Therefore, the firm may find it profitable to discontinue some of the non-subsidized projects, thereby contributing to a reduction in privately financed R&D expenditures.

In sum, an R&D subsidy can have a direct and an indirect effect on a firm's performance. The direct effect operates through the increase in total R&D expenditures, holding privately financed R&D constant. The indirect effect operates through the response of privately financed R&D expenditures to the subsidy. If the R&D subsidy displaces a firm's own R&D expenditures, the total effect may be modest or even negative. On the other hand, if it stimulates a firm's own R&D expenditures, then the effects of the subsidy are magnified. Thus, an understanding of the relationship between R&D subsidies and privately financed R&D is necessary for a correct assessment of the role of R&D subsidies in boosting privately financed R&D expenditures.

Firms may react differently to the R&D subsidy depending, essentially, on their own characteristics and on the features of the subsidy. Firms facing better markets or higher probabilities of R&D success, for example, will have larger R&D departments and hence more R&D expenditure. The same is true for firms with lower costs of setting up R&D projects. These firms will be more technologically advanced and, even controlling for the technological level, will be more valuable. According to the classical result of Dasgupta and Stiglitz (1980) (see also Lee, 2002), under certain conditions a firm's R&D spending as a share of its sales revenue (R&D effort) depends positively on the R&D elasticity of demand and inversely on the price elasticity of demand. In this context, several firm and industry characteristics would shape different relationships between these variables.

Furthermore, the characteristics of a subsidy may also have an impact on its specific effects on privately funded R&D expenditures. In this case, both the amount of the subsidy as well as the intensity may have different effects depending on the features of the recipient firm. Moreover, due to the intricate relationship between the different characteristics of firms and subsidies, the reaction of a firm's R&D effort to the subsidy's intensity is expected to be non-linear. From the previous analysis, it would appear that whether R&D subsidies stimulate or crowd out private R&D expenditures is, therefore, an empirical matter.¹

The empirical literature evaluating public R&D subsidies has become increasingly devoted to analysing possible heterogeneous effects depending on the characteristics of firms and the subsidies. Apart from analysing the full impact of R&D subsidies on recipient firms, knowing whether the effects are heterogeneous across firms provides relevant information for the design of technology policy.

Recent studies have taken into account a number of characteristics of the firm, including size, sector and location, in examining possible differences in the effects of R&D subsidies (Lach, 2002; Czarnitizki and Fier, 2002; Czarnitzki, 2006; González et al.,

¹ Note also that we restrict ourselves to the effect of the subsidy on a firm's own R&D expenditures. Subsidies may also have implications for other non-R&D activities, both contemporaneously and over time, and, through inter-firm spillovers or rivalry channels, subsidies may have effects on other firms' R&D activities.

González and Pazó 2008, Cerulli and Poti, 2010). The main conclusion of these analyses is that the effects of R&D subsidies are not homogenous. For instance, for Spanish manufacturing firms, González and Pazó (2008) examine the differences in the effectiveness of subsidies according to firm size and industry type and conclude that public support is more effective in small firms and firms operating in low-technology sectors. In addition, other recent analyses have sought to determine whether the effects of public subsidies differ depending on their specific characteristics. For instance, using countries as their unit of analysis, Guellec and van Pottelsberghe (2003) point out that the effectiveness of government support seems to have an inverted U-shape, increasing up to a particular threshold and decreasing after that. Busom (2000) suggests that a small subsidy does not modify the scope of a firm's R&D plans, but admits that without knowing the exact amount of the subsidy it is impossible to confirm this hypothesis. González et al. (2005) show that subsidies induce modest increases in privately financed effort and that the impact grows, albeit only in a small proportion, with the size of the subsidy. They also estimate the amount of the subsidy, as a percentage of a firm's R&D expenditures, needed to induce non-performing firms to perform innovative activities.

Recent contributions also show that the effects of public subsidies may vary with their size and intensity. Specifically, Görg and Strobl (2007) use three different subsidy size categories (small, medium and large) and examine their effects for domestic plants and multinationals in Ireland. In the case of domestic plants, small- and medium-scale subsidies do not crowd out private R&D spending, while some additional effects are found for small subsidies, and a crowding out effect is reported for large-scale subsidies. For foreign multinationals, the effect does not vary with subsidy size. Aschhoff (2009) shows for Germany that small subsidies do not have an effect on privately financed R&D expenditures, whereas medium and large subsidies do have additional effects. Finally, Cerulli and Poti (2010) consider three classes of financing intensity, defined as the subsidy share of the total project cost, and their results show that the positive effect of the subsidy only appears above a certain threshold, when about 50% of the total project cost is being financed.

In short, on the basis of both the theoretical considerations and the empirical analyses, it can be concluded that the effects of R&D subsidies are not homogenous. Identifying

possible differences, depending on the characteristics of the subsidies, would ensure an improvement in the effectiveness of public support to private R&D. In particular, the intensity of the subsidy is one of the main decisions to be taken by public agencies, and knowing how firms react to different financing intensities would provide essential information for a better understanding of this policy, and lead to an improvement in the impact and efficiency of public programmes.

3. Empirical strategy

Given the nature of our goal, i.e., analysing the optimal percentage of R&D subsidy (or R&D intensity), we draw on all relevant econometric techniques that might help us understand, as identified by recent empirical analyses and theoretical models, the possible non-linear relationship between the percentage of R&D subsidy granted by the public agency and the private R&D effort made by recipient firms.

As such, our empirical strategy is two-fold: a parametric and a non-parametric set-up. The former seeks to compare our results with those reported in the R&D literature. Here, we use a well-established methodology which also allows us to control for well-known problems including selection bias and endogeneity. The non-parametric set up, a more flexible framework, adds important information and evidence.

3.1 Parametric set-up: two-step procedure

The main equation to estimate relates a given percentage of an R&D subsidy as a determinant of a firm's private R&D effort, or in formal terms,

$$RDe = f(\% RDi, X_i),$$
[1]

where *RDe* is the private R&D effort of a recipient firm measured as private R&D expenditure as a percentage of sales; %*RDi* is the R&D subsidy intensity measured as the percentage (on eligible cost) of the R&D subsidized by the public agency and X_i is a set of firm characteristics (see section 4 for more details on data issues and the definition of variables). By estimating Eq. [1], we aim to explain whether receiving a given percentage of subsidy to perform R&D activities has any impact on a firm's R&D

effort. Thus, we do not seek to determine whether receiving a public subsidy has an impact on R&D effort, as has been usual in the literature to date. Rather, we go a step further and ask whether the specific percentage received has an impact on the R&D effort of private firms.

Two important issues must be taken into account when estimating Eq.[1] econometrically: a) possible problems of endogeneity of the actual intensity of the R&D subsidy and b) possible problems of selection bias. Below, we deal with these two potential econometric problems.

a) First step: endogeneity

Our first concern is that our main variable of interest, *%RDi*, could be endogenously determined, that is, the percentage of private R&D activities subsidized publicly could determine the effort in R&D made by recipient firms; but it could also be the case that firms with a higher R&D effort receive a higher percentage of public subsidy precisely because the effort they dedicate to R&D is higher. Therefore, we need to control for the likely existence of endogeneity in Eq. [1]. To deal with this, we initially estimate the determinants of R&D subsidy intensity as follows:

$$\% RDi = f(X_i, P_i)$$
^[2]

where X_i is again a vector of variables including firm characteristics and P_i are the scores obtained by the R&D projects presented by firms to obtain public subsidies (section 4 explains the nature of this variable in more detail). Therefore, Eq. [2] presents the determinants of receiving a specific percentage of the total subsidy applied to the public agency and allows us to study the role of the scores given to private R&D projects in deciding the subsidy intensity. In this sense, we are able to analyse the granting decision of the public agency in detail.

Note that Eq. [2] must be estimated using a Tobit model given the distribution of *%RDi*. This variable has a specific distribution: for non-recipient firms it is 0, while for recipient firms it takes a positive value between 0 and an upper bound of 60% according to EU criteria for subsidy intensity. Moreover, the estimation of Eq. [2] does not suffer

from selection bias because this variable only exists for firms that apply for a public subsidy. The percentage of publicly subsidized R&D is at the total discretion of the public agency granting the subsidies and, hence, this variable exists only for firms that apply for subsidies and, in our case, we account for all of them.

We use the estimation of Eq. [2] to obtain the predicted values for % RDi, that is, *pred*[% RDi]. If this first stage is well specified, the predicted values can be used in the second stage to tackle the possible problem of endogeneity outlined above. Therefore, we can re-estimate Eq. [1] as follows:

$$RDe = f(pred[\%RDi], X_i)$$
[3]

b) Second step: selection bias

The estimation of Eq. [3], i.e., the determinants of the firms' R&D effort, may likewise suffer from selection bias. The possible existence of this bias (which has been well established in the empirical literature dealing with R&D subsidies) is attributable to the fact that all firms may have a certain R&D effort, but we are only able to observe the R&D effort of those firms that apply for a public subsidy. Therefore, extending our results to all firms would be misleading, since we are observing the R&D effort of a particular sample of firms that could, *a priori*, have a higher R&D effort than the rest of firms.

Hence, we need to estimate Eq. [3] by taking into account the possible existence of this selection bias. Various econometric techniques allow us to perform statistical tests to verify if this selection bias exists. In line with the literature (see, for instance, Hussinger, 2008), we estimate Eq. [3] using the Heckman procedure:

$$RDe = f(pred[\%RDi], X_i, \lambda)$$
[4]

In Eq. [3] we introduce the Inverse Mills Ratio (λ) derived from the estimation of a selection equation (binary participation decision) that models the probability of receiving a public subsidy, using a probit model. More formally the selection equation is:

$$Receive = f(X_i, Y_i)$$
^[5]

where *Receive* is a dummy variable indicating the receipt of a public R&D subsidy; X_i are the same variables as before while Y_i corresponds to the characteristics of the project that could affect the probability of receiving a public subsidy. Under normality the coefficient of λ in Eq. [4] is proportional to the hazard rates and depends only on known parameters of Eq. [5].

Hence, combining Eq. [5] and Eq. [4] we perform a Heckman estimation, which in our case accounts for the possible problems of selection bias as well as those of endogeneity (since we use the predicted values obtained in the first stage).

3.2. Non-Parametric set-up: conditional quantile regression approach

In our context, the assumption of linearity in the relationship between public R&D subsidy intensity and the firms' private R&D effort may not be appropriate: the issue of specifying the functional form of the model emerges, and this is a particularly difficult choice especially when the impact of the regressors on the regressand is not well established by a theoretical model, as in our case.

For this reason, we avoid any a priori assumptions about the specific functional form, and use a complete non-parametric estimator. By so doing, we obtain a graphical idea of the relationship existing between the variables under examination.

The joint cumulative density function F(% RDi, RDe), assuming that the bivariate distribution f(% RDi, RDe) exists, is given by:

$$F(RDe|\%RDi) = \int_{-\infty}^{RDi} f(RDe|\%RDi) dRDe = \int_{-\infty}^{RDi} \frac{f(\%RDi, RDe)}{f(\%RDi)} dRDe \qquad [6]$$

where

$$f(\% RDi) = \int_{-\infty}^{\infty} f(\% RDi, RDe) dRDe$$
[7]

represents the marginal distribution of %RDi.

The inverse function $F^{-1}(p|\%RDi)$ of Eq. [6] gives the *p*-quantile (parametrically estimated) of *RDe* conditional on %RDi. The problem is that the functional form of *f*(.) is unknown and must be estimated. Instead of using a priori assumptions about its shape, we estimate it non-parametrically (Trede, 1998). Different non-parametric approaches have been used in the literature, such as spline smoothing (Koenker, Portnoy and Ng, 1992) and kernel density estimation (Abberger, 1997). We use the latter because we consider the idea that the shape of f(%RDi, RDe) in Eq. [6] is unknown and has to be replaced by its non-parametric estimate (Trede 1998) intuitively appealing. Let *n* be the number of observations for which the private R&D effort and the public R&D subsidy intensity are available, rde_i and rdi_i being the measured variables relative to the *ith* firm of the sample, and i = 1, ..., N. The bivariate density can be defined as:

$$\hat{f}(rdi, rde) = \frac{1}{nh_1h_2} \sum_{i=1}^n K\left(\frac{rdi - rds_i}{h_1}\right) K\left(\frac{rde - rde_i}{h_2}\right)$$
[8]

where h_1 and h_2 are the bandwidths, and K(.) is the Kernel.² Then substitute this expression into Eq. [6], setting the cumulative density function of the Kernel equal to:

$$H(z) = \int_{-\infty}^{z} K(x) \delta x$$
[9]

we obtain:

²More generally, a multivariate density with dimension d may be estimated as:

$$\hat{f}_{H}(x) = \frac{1}{n} \sum_{i=1}^{n} k_{H}(x - X_{i})$$

$$k(u) = \prod_{j=1}^{a} K(u_j),$$

where K(.) is the univariate Kernel function.

where k(.) is a *d*-dimensional Kernel function, and *H* is a bandwidths matrix. Setting in such function d=2 gives the bivariate density function. "A convenient choice in practice is to take $H=hS^{0.5}$, where *S* is the sample covariance matrix and *h* is a scalar bandwidth sequence, and to give *k* a product structure" (Härdle and Linton, 1994). In other words, with respect to the univariate case, in the multivariate setting the issue of choosing a relation between kernels arises, and it is usually solved by using a multiplicative structure:

$$\hat{F}(rde|rdi) = \frac{\sum_{i=1}^{n} K\left(\frac{rdi - rdi_i}{h_1}\right) H\left(\frac{rde - rde_i}{h_2}\right)}{\sum_{i=1}^{n} K\left(\frac{rdi - rdi_i}{h_1}\right)}$$
[10]

that represents the conditional density function of *rde* to *rdi*. By inverting equation Eq. [10] we obtain the non-parametric quantile *p* that we are looking for, which depends on the percentage of R&D subsidy received by firms. Again, by choosing $p \in [0,1]$, an infinite number of quantiles of the relationship under investigation can be obtained; by a similar procedure (Local Least Squares) it is also possible to derive a non-parametric version of the mean function (Härdle, 1990). Interestingly for our purposes, this method allows us to estimate the relation of %*RDi* for different types of firms, in this case, types defined by the distribution of the private R&D effort (*RDe*).

Unlike the parametric analysis, the non-parametric approach is bivariate: the nonparametric conditional quantile gives the effect of the percentage of R&D subsidy on the firms' R&D effort, together with the effect of variables that may possibly influence this relationship.

4. Data

In order to proceed with the empirical strategy proposed above, we need data on the characteristics of the applicant firms and their R&D projects, the scores awarded by the agency to each project and, finally, data on the characteristics of the subsidy granted. For this purpose, we draw on a complete database, which covers all the relevant information needed, for a sample of 968 firms that applied for a public R&D subsidy in Catalonia (Spain) in the years 2005 and 2006.

Catalonia has one of the largest regional innovation systems in Spain, concentrating around 26% of private R&D expenditure. Its total R&D expenditure represents 1.6% of regional GDP. Moreover, it was one of the first regions to pursue an independent R&D subsidies policy in Spain. In Catalonia, R&D subsidies are granted by the Catalan public agency in a process similar to international procedures (see Jaffe, 2002 and Tanayama, 2009). The procedure is as follows. First, the agency allocates an overall

budget to the call for R&D subsidies. Second, firms apply for subsidies by submitting a proposal for an R&D project. This application contains qualitative and quantitative information on both the firm and the R&D project itself. Third, in accordance with established (and public) criteria, experts evaluate the projects and award a numerical score that allows all projects presented to be ranked. Finally, a committee appointed by the agency uses this information to determine which projects will be granted a subsidy and the intensity of that subsidy. Therefore, by drawing on this complete database, we have access to all the information required to estimate the optimal subsidy applying the methodology described above.

Scores

As explained above, in order to overcome the possible problem of endogeneity in the estimation of the determinants of public R&D subsidy intensity we use, as an instrument, the scores (P_i) received by each private R&D project submitted. The agency publishes in their calls for firms' R&D projects the criteria used in the evaluation and selection process. These criteria are related to the four main objectives of the agency and (internal and external) referees are asked to evaluate each project accordingly. The four variables used are:

i) Technological contribution (**tech_contrib**): refers to the agency's goal to promote technologically outstanding projects.

ii) Socio-economic impact (**se_impact**): refers to the assessment of the impact of projects in terms of their Gross Value Added and/or R&D investment.

iii) Internationalization (**international**): refers to the estimated contribution of the project to the internationalization of firms.

iv) Diffusion (**diff**): takes into account the impact of the project on other sectors and on other firms.

In addition to these four variables, which capture the extent to which the private R&D projects fulfill the agency's four main objectives, we also use, as a robustness exercise, the overall score obtained by each project (**total**).

A priori these variables should be highly correlated with the percentage of the subsidy eventually granted, but they could be considered as being independent of the firms' private R&D effort, thereby making them a good instrument for tackling the possible problem of endogeneity in the first stage of the estimations.³ Interestingly, and unlike in most previous studies evaluating R&D subsidies, this information allows us to examine the agency's allocation procedures because, as a number of recent analyses show (Takalo *et al.*, 2008, Huergo and Trenado, 2010), project-level characteristics are more important than firm-level characteristics in the decision to grant a public subsidy.

Characteristics of firms, projects and subsidies

As mentioned above, we have data on different characteristics of both the firms and the projects submitted to the public agency. Applications to obtain a public subsidy in Catalonia require the firms to complete a questionnaire regarding both the proposed R&D project and their own characteristics. More specifically, firms must report past, current and projections of their main economic variables. This means we are able to use past values of some variables of interest in our econometric estimations, as is common practice in the literature, and thus avoid the problems of endogeneity of some contemporaneous regressors to private R&D effort. Table 1 contains descriptions of the explanatory variables used in the empirical estimations.

Firm-level characteristics are mainly used to control for firm specific factors that may affect decisions about the concession and intensity of subsidies. They include such characteristics as age, size, location and sector, as well as R&D and export activities. Information about the firm's application is also included, for example, whether the firm has applied for several projects in one year and whether the firm applied in the two years included in our sample. Given the cross-sectional approach, this information is

³ For instance, in our case, scores are determined at the project level, while private R&D effort is given at the firm level. A simple partial correlation between project scores and firm R&D effort shows results below 10%.

likely to capture some of the unobserved heterogeneity across firms related to their innovative activities.⁴

<INSERT TABLE 1 AROUND HERE>

Table 2 describes the available data. From this table we observe that firms that were granted a subsidy for their R&D projects tended to be larger in terms of their number of employees, to be older, to have greater export intensity and, in terms of R&D activities, to have registered more patents and, on average, to have higher R&D expenditure. Moreover, R&D projects that have applied for a public subsidy account for a large proportion of the firms' private R&D expenditure, with an average percentage of the cost of the R&D project of almost 70% of firms' private R&D expenditure.

As for the variables used in the project evaluation process, the main two criteria employed by the agency were, first, the technological content of the proposal (with a range from 0 to 175) and, second, its expected socio-economic impact (with a range from 0 to 125). This suggests that the agency seeks to select the best projects in terms of their technological potential while taking into consideration their effects on the whole economy. In addition, the agency seems to adopt a "picking-winners" strategy, thus promoting projects that develop products which can be introduced in the market and which can contribute to the internationalisation of the firm. Finally, diffusion effects and possible intra- and inter-firm spillovers are also considered by including among the criteria the effects on other firms and on other sectors. Yet, the scores recorded by firms in this last criterion are, on average, quite low. The descriptive statistics show, as expected, that successful projects receive higher scores on average for all the items considered. However, the scores differ substantially in the criteria corresponding to the expected effect of private R&D projects on the internationalisation of the firms and on their socio-economic impact.

<INSERT TABLE 2 AROUND HERE>

5. Results and discussion

⁴ Unfortunately, we were unable to obtain information as to whether the firms were participating in other public calls for R&D subsidies (at the Spanish or European level).

5.1 Parametric results

Table 3 shows the results for the estimation of the Tobit model, Eq. [2], in the first stage of the two-step procedure. This estimation concerns the public agency's rule of subsidizing a specific percentage of the awardable cost of a successful applicant's R&D project (%RDi), i.e., the public R&D subsidy intensity. The set of independent variables are firm-level and project-level characteristics. The selection of firm characteristics is in line with the literature (Busom, 2000; Lach, 2002; Czarnitizki and Fier, 2002; Czarnitzki, 2006; González and Pazó 2008, among others) and includes both past and current indicators of the technological activity of the firms such as the existence of an R&D department (**RDdept**), the number of patents (**patents**_{t-1}), the export intensity of the firm $(exports_{t-1})$ and the percentage of private R&D subcontracted to other firms and public research centres (subcontract_ R_{t-1}). We also include variables to capture other firm characteristics, such as, size (emp), sector (sht and dht) and age, that may influence the selection of the R&D projects by the agency and the percentage of public funds received. This set of variables also includes a dummy to control for whether the firm applied for a subsidy in the previous period (repeat), thereby taking into account the persistence of R&D activities and the experience of the firm in applying for public subsidies.

In addition, to estimate the probability of a firm receiving a specific subsidy intensity, we complemented this set of firm characteristics with project-level scores (given their influence on the agency's decision) in a similar procedure to that reported in Takalo *et al.* (2008). As explained above, we used the total score received by R&D projects (**total**) and the four main criteria used in the evaluation of these projects: technological contribution (**tech_contrib**), economic impact (**se_impact**), effects on internationalisation (**international**) and the diffusion of the results and cooperation with other agents (**diff**).

<INSERT TABLE 3 AROUND HERE>

Note from Table 3 the importance of taking into account the project scores when analysing the agency decision, a procedure adopted in only a very few recent empirical

studies (Takalo *et al.*, 2008; Huergo and Trenado, 2010). Our results show that the decision making of the Catalan agency is in line with their publicly stated objectives and funding principles. The scores awarded to the projects have a greater influence on the agency's decision regarding the intensity of the subsidy than do the specific characteristics of the firms. The four criteria used by the agency to evaluate the firms' R&D projects are statistically significant. Among these, the technological content of the projects and their estimated contribution to the internationalisation of the firms have the greatest effect on subsidy intensity.

Furthermore, our results regarding firm-level characteristics are consistent with the literature. Thus, having previous experience in R&D projects with public support has a positive impact on the intensity of the subsidy. The results indicate that the agency prefers to give a higher subsidy to firms that acquire external research, in most instances from universities or public research centres. Except in the case of age (the public agency shows a higher propensity to support the projects of well-consolidated firms) and size (the agency seems to favour smaller firms), the rest of the variables are not significant. Specifically, the location of firms operating in a high technology sector does not influence the agency's decision regarding the intensity of the subsidy.

The Tobit estimations of Eq. [2] presented in Table 3 allow us to calculate the predicted values of the subsidy intensity (pred[%RDi]) and to use them in the second-stage estimation of the R&D effort equation (see Eq.[4] and Eq.[5]). As explained above, this second stage is performed using the Heckman two-step procedure to avoid possible problems of selection bias. Nevertheless, we also perform OLS estimates to compare both sets of results. The results are presented in Tables 4 and 5.

<INSERT TABLE 4 AROUND HERE>

In the estimation of the selection equation, Eq.[5], i.e., the first-step in the Heckman procedure, and in line with the literature, we include a number of indicators of the firms' technological activity and other firm characteristics such as size, sector and previous private R&D activities. This estimation provides further information regarding the agency's decision to allocate subsidies and shows that previous private R&D spending, location outside the metropolitan area of Barcelona, and previous

participation in a call for public R&D subsidies positively affect the propensity to receive a subsidy. Therefore, the agency seems to prefer supporting firms with the capacity to undertake R&D activities. Our results also show the need to include a variable (**repeat**) that takes into account the persistence of a firm in applying for public subsidies (González and Pazó, 2008; Hussinger, 2008). These results suggest that the strategy of the agency is to fund the most promising candidates with experience in R&D activities. Nevertheless, since most R&D activities are concentrated in the Barcelona metropolitan area, the agency does follow certain criteria of regional distribution by favouring R&D activities outside Barcelona.

In the estimation of the impact of subsidy intensity (% RDi) on firms' R&D effort (*RDe*), using both OLS and the Heckman procedure, we introduce both the linear predicted value of % RDi and the square of the predicted value of % RDi, that is, we allow our estimates to account for the possible existence of nonlinearities in our main relation of interest.

<INSERT TABLE 5 AROUND HERE>

The results in Table 5, controlling for the variables that may affect the firms' private R&D effort, show a positive, albeit non-significant effect of public R&D subsidy intensity. In the estimates using the Heckman procedure, a priori more reliable given that the Mills ratio is statistically significant, the square term of public R&D subsidy intensity is negative (again not significant), pointing to the possible existence of an inverted U-shape in the relationship between subsidy intensity and the private R&D effort.

The effects associated with the main variables accounting for the firms' R&D effort are in line with recent analyses conducted for the Spanish and Catalan economies (Herrera and Heijs, 2007; Duch *et al.*, 2009). Specifically, our results show that the firms' R&D effort is related positively with performing R&D systematically, having more R&D employees, being located in the metropolitan area of Barcelona and, finally, belonging to sectors classified as knowledge intensive services. In addition, younger and smaller innovative firms make a greater R&D effort measured as a percentage of their turnover. In short, from the whole parametric set-up we can highlight the following results. The Tobit stage shows, as expected, the key role played by the scores awarded by the public agency to the R&D projects presented for public subsidies. This variable has a high explanatory potential of the percentage of R&D subsidy eventually granted and allows us to avoid possible problems of endogeneity (of public R&D subsidy intensity with regards to private R&D effort) in the next step of the estimation procedure. Hence, in the estimation of the determinants of firms' R&D effort, using the well-known two-step Heckman procedure, we find that the Mills ratio is significant, which indicates that there is selection bias in our sample. Moreover, the first step in this procedure allows us to account for the determinants of a firm being granted a public subsidy and to compare our results with those in the literature. Having controlled for endogeneity and selection bias, we can then estimate the equation explaining the determinants of the private R&D effort. In this case, our variable of interest [%RDi] is positive and its square (which accounts for possible nonlinearities in the established relation) negative, both being non-significant. Therefore, the results of the estimations, although not conclusive, seem to indicate a possible inverted U-shape. This means that, on average, R&D subsidy intensity could foster private R&D intensity, although this impact may not be equal for all firms and may depend on the R&D subsidy intensity itself. Moreover, the parametric estimates indicate the need to examine in more detail the existence of a non-linear relationship between subsidy intensity and the firms' R&D effort.

5.2 Non-Parametric results

Given that we cannot take the parametric set-up any further, we shift the analysis to a more flexible framework to capture a possible non-linear relationship between the percentage of subsidy awarded and the R&D effort made by recipient firms. Figure 1 presents a basic non-parametric quantile regression of private R&D effort on public R&D subsidy. It seems that there is a positive relationship between subsidy intensity (%*RDi*) and private R&D effort. Nevertheless this simple relationship suffers from the possible endogeneity problem.

<INSERT FIGURE 1 AROUND HERE>

As in the previous framework, if we use the Tobit predicted values of *%RDi* to control for endogeneity, we obtain a different picture. Figure 2 shows the relationship between public R&D intensity and private R&D effort when controlling for endogeneity in the whole distribution of private R&D effort. Note that there would only appear to be an impact of the public R&D subsidy intensity for high quantiles (90%) of the private R&D effort distribution.

<INSERT FIGURE 2 AROUND HERE>

The non-parametric evidence seems consistent with the parametric evidence reported above and provides more detailed information on the effects of R&D subsidy intensity. The results of the estimations in the parametric framework pointed to the possibility of a non-linear effect of subsidies on firms' R&D effort, although this effect was not statistically significant. This non-significance is now well accounted for by the fact that the impact of R&D subsidy intensity (%RDi) on private R&D effort for firms located at the mean and in the low quantiles of the private R&D effort distribution is close to zero. However, there is an impact of the subsidy intensity for those firms with high R&D effort.

These results point to the need to take the analysis beyond that of the average effects of R&D subsidies when evaluating public R&D programmes. Public support to firms' R&D projects, aimed at encouraging the firms to allocate more of their own resources to R&D activities, reduces the marginal costs and increases the profitability of these projects. On the one hand, our results indicate that firms with a low R&D effort tend to undertake projects that are sufficiently interesting to them and which they can finance themselves. In these circumstances, public support has no additional effect on a firm's R&D effort. On the other hand, firms with a higher R&D effort are able to undertake new projects, probably ones with lower expected returns, and to increase their R&D if they receive public support. Nevertheless this public support needs to reach a certain threshold in order to be effective and stimulate new privately financed R&D.

For the most R&D intensive firms, the relationship between R&D effort and the intensity of the R&D subsidy is highly non-linear⁵. The specific shape of the relationship depends on the intensity of the R&D subsidy and we have identified two clearly separated sets. In the first, corresponding to firms with a low level of R&D subsidy intensity, the relationship takes a U-shaped form. In the second, corresponding to firms with a high level of R&D subsidy intensity, we find that the relationship adopts an inverted-U shape. Overall, as the intensity of the R&D subsidy increases up to a point around 11%, R&D effort declines. From that value up to an R&D subsidy intensity of around 42%, R&D effort increases steadily before falling again. Thus, there exists an optimal percentage for the public subsidy in terms of maximizing firms' R&D effort.

According to Lee (2002, 2011), a firm's profit-maximizing R&D effort is determined by the interaction of consumer characteristics, represented by the elasticity with respect to price (quality), and firm-specific technological competence (or R&D productivity) measured as the R&D elasticity or technological output. As we have seen in section 2, an R&D subsidy can affect both variables, but the intensity of the effect is likely to vary with firm and subsidy characteristics. Considering the fact that all firms in the same industry face almost the same consumer preferences, technological competence plays a key role in determining a firm's R&D effort. Our results show that for extreme values of the distribution of the R&D subsidy intensity the R&D elasticity of demand (or the firm's R&D productivity) declines. In these cases, the indirect effect of the R&D subsidy displaces some R&D expenditure and R&D effort falls. On the other hand, for intermediate values of intensity distribution, R&D effort increases as a consequence of a rise in R&D productivity. At these values, the effects of the R&D subsidy stimulate privately financed R&D.

6. Conclusions

Public subsidies constitute an essential tool of any technology policy given their role in overcoming market failures and ensuring that the allocation of private resources to R&D

⁵ Note that the relationships underlined in the non-parametric framework cannot be captured with the parametric regressions that only show the sign and significance for the linear term (positive) and for the square term (negative).

activities reaches the social optimum. In designing technology policy, the question as to whether public support complements or substitutes privately financed R&D is fundamental and, as such, has been the subject of numerous empirical analyses. However, most studies evaluating R&D subsidies tend to analyse only the average effects of these subsidies on recipient firms, despite the fact that some recent analyses have shown that the degree of effectiveness of R&D policy may differ substantially according to allocation procedures and the characteristics of the R&D subsidies themselves. Among the latter, the amount and intensity of the subsidy are particularly relevant characteristics for the design of public schemes in support of private R&D.

This paper contributes to the empirical literature on the evaluation of R&D policy by examining whether the magnitude of the effects of R&D subsidies depend on the intensity of this public support. In addition, firm and project characteristics have been included to examine the process followed by public agencies in their allocation of R&D subsidies. To carry out the analysis, parametric and non-parametric econometric techniques have been used. In the econometric procedure, the two main problems that an analysis of this type must face - endogeneity and selection bias - have been taken into account. The main results of the econometric analysis, and the conclusions that can be drawn from it, are the following.

Firstly, our knowledge regarding how public R&D subsidies are allocated by agencies remains insufficient. Our empirical analysis shows the importance of taking into account project-level characteristics (specifically, the scores awarded to them) in any examination of the allocation of R&D subsidies by the public agency. These characteristics appear to have a stronger influence than firm-level characteristics, which to date have served as the common explanatory variables in the empirical literature. As such, information about a project is particularly relevant for strengthening our understanding of the objectives of the public agency and its procedures for awarding R&D subsidies. Our results show that the granting agency's decision making is consistent with its publicly stated objectives. Subsidies are oriented mainly to promote projects with a high technological content, while considering their impact on the analysis allows us to tackle the possible problem of endogeneity when relating private R&D effort with public subsidy R&D intensity.

Secondly, for a better understanding and evaluation of R&D subsidies we need a fuller picture than that provided by their average impact on recipient firms, since the effects are heterogeneous depending on the specific characteristics of the firms and the subsidies themselves. Our results show that the effect of subsidy intensity on a firm's R&D effort seems to be non-linear and that while, on average, the subsidies have no effect, their impact acquires relevance according to the firm's R&D effort and the subsidy intensity. In the case of less R&D intensive firms, the impact of subsidy intensity on private R&D effort is close to zero. However, for firms with high R&D effort, subsidy intensity has a positive impact. For this latter group, the relationship between R&D effort and the intensity of the R&D subsidy is highly non-linear, with the actual shape of the relationship depending on the intensity of the R&D subsidy at which the R&D effort of firms is maximized.

Our results provide information for evaluating R&D subsidies and have implications for the design of R&D subsidy programmes, specifically, and technology policy, in general. Determining the intensity of such subsidies is one of the key issues that public agencies must respond to as part of the process of granting subsidies. Our results show that only after a certain threshold are subsidies effective. Furthermore, we show that in the case of firms with a low R&D effort, subsidies may not be sufficient to increase their R&D effort.

Our analysis is not free of limitations. Although our results show that firms with greater R&D effort are able to develop new projects - albeit probably with lower expected private returns - the fact that they receive public support is insufficient to conclude that these projects are the ones of greatest social interest. In this sense, and although financial additionality is important and a necessary condition, a comprehensive *ex-post* evaluation would need to take into account the whole set of effects of R&D subsidies and, in particular, the spillovers generated.

References

- Abberger, K. (1997): "Quantile smoothing in financial time series", *Statistical Papers* 38.
- Aschhoff, B. (2009): "The effect of subsidies on R&D investment and success Do subsidy history and size matter?", Discussion Paper 09-032, ZEW
- Arrow, K. (1962): "Economic welfare and the allocation of resources for inventions" in R. Nelson (ed.): *The rate and direction of inventive activity*, Princeton University Press.
- Busom, I. (2000): "An empirical evaluation of R&D subsidies", *Economics of innovation and new technology*, 9(2), 111-148.
- Cerulli, G., and Poti, B. (2010): "The differential impact of privately and publicly funded R&D on R&D investment and innovation: The Italian case", Working Papers 10, Doctoral School of Economics, Sapienza University of Rome, revised 2010.
- Czarnitzki, D. and Fier, A. (2002): "Do innovation subsidies crowd out private investment: evidence from the German service sector", *Applied Economics Quarterly*, 48(1), 1–25.
- Czarnitzki D. (2006): "Research and development in small and medium-sized German enterprises: The role of financial constraints and public funding", *Scottish Journal of Political Economy*, 53 (3), 335-357.
- Dasgupta, P., and Stiglitz, J. (1980): "Industrial structure and the nature of innovative activity", *Economic Journal*, 90, 266-293.
- David, P., Hall, B. and Toole, A. (2000): "Is public R&D a complement or substitute for private R&D? A review of the econometric evidence, *Research Policy*, 29, 497-529.
- Duch, N., Montolio, D., and Mediavilla, M. (2009): "Evaluating the impact of public subsidies on firm's performance: a two-stage quasi-experimental approach" *Investigaciones Regionales* 16, 143-165.
- García-Quevedo, J. (2004): "Do public subsidies complement business R&D? A metaanalysis of the econometric evidence", *Kyklos*, 57(1), 87-102.
- González, X., Jaumandreu, J., and Pazó, C. (2005): "Barriers to Innovation and Subsidy Effectiveness", *Rand Journal of Economics*, 36(4), 930-949.
- González, X. and Pazó C. (2008): "Do public subsidies stimulate private R&D activities?", *Research Policy*, 37(3), 371-389.
- Görg, H. and Strobl, E. (2007): "The effect of R&D subsidies on private R&D", *Economica* 74, 215-234.
- Guellec, D. and Van Pottelsberghe, B. (2003): "The impact of public R&D expenditure on business R&D", *Economics of innovation and new technology*, 12(3), 225-246.
- Härdle, W. (1990): Applied nonparametric regression, Cambridge University Press, New York.
- Härdle, W. and Linton, O. (1994): "Applied Nonparametric Methods" in *Handbook of Econometrics*. Heckman and Leamer Eds. Vol 4. North-Holland.
- Heckman, J. (1979): "Sample selection bias as a specification error". *Econometrica*, 47, 153-61.
- Herrera, L. and Heijs, J. (2007): "Difusión y adicionalidad de las ayudas públicas a la innovación". *Revista de Economía Aplicada*, 44, 177-197.

- Huergo, E. and Trenado, M. (2010): "The Application for and the Awarding of Low-Interest Credits to Finance R&D Projects". *Review of Industrial Organization*, 37(3), 237-259.
- Hussinger, K. (2008): "R&D and subsidies at the firm level: an application of parametric and semiparametric two-step selection models". *Journal of Applied Econometrics*, 23, 729-747.
- Jaffe, A. (2002): "Building programme evaluation into the design of public researchsupport programmes". *Oxford Review of Economic Policy*, 18, 22–34.
- Koenker, R., Portnoy, S. and Ng, P. (1992): "Nonparametric estimation of conditional quantile functions", in Dodge, Y. (Ed), *L₁-Statistical Analysis and Related Methods*. North-Holland, New York.
- Lach, S. (2002): "Do R&D subsidies stimulate or displace private R&D? Evidence from Israel", *Journal of industrial economics*, 50(4), 369-390.
- Lee, Chang-Yang (2002) "A Simple Model of R&D: An Extension of the Dorfman– Steiner Theorem", *Applied Economics Letters*, **9**, 449–452.
- Lee, Chang-Yang (2011) "The differential effects of public R&D support on firm R&D: Theory and evidence from multi-country data", *Technovation*, **31**, 256-269.
- Takalo, T., Tanayama, T., and Toivanen, T. (2005): "Selection or self-rejection? Applications into a voluntary treatment program: The case of R&D subsidies", *Helsinki Center of Economic Research HECER Discussion Papers* 76.
- Takalo, T., Tanayama, T., and Toivanen, T. (2008): *Evaluating innovation policy: a structural treatment effect model of R&D subsidies*, Bank of Finland Research Discussion Papers 7, 2008.
- Tanayama, T. (2009): "Rationales and reality of R&D subsidies: Are SMEs and large firms treated differently?", *Helsinki Center of Economic Research HECER Discussion Papers* 280.
- Toivanen, O. (2006): "Innovation and research policies: two case studies of R&D subsidies", EIB Papers 11(2), 54-78.
- Trede, M. (1998): "Making mobility visible: a graphical device", Economic Letters, 59.
- Wallsten, S. (2000): "The effects of government-industry R&D programs on private R&D: the case of the Small Business Innovation Research Program", *Rand Journal of Economics* 31, 82-100.

Tabl	le 1	. D	escrit	otion	of 1	the	exp	lanat	torv	varia	bl	les
		• -			· · ·		p		· · · · ·		~ .	

Variable (abbreviation)	Description
Firm-level	
Size (emp)	Number of employees
Age (age)	Number of years the firm has been operating
Location (dloc)	Dummy indicating whether the firm is located in Barcelona
High technology manufacturing (dht)	Dummy indicating whether the firm is in a high technology manufacturing sector
Knowledge intensive services (sht)	Dummy indicating whether the firm is in a knowledge intensive service sector
Export intensity (exports)	Exports as a percentage of sales
Number of patents (patents)	Number of patents registered by the firm
Availability of an R&D department (RDdept)	Dummy variable indicating whether the firm has an R&D department
R&D employment (RDemployment)	Number of employees dedicated exclusively to R&D activities
R&D expenditure (RDexpenditure)	R&D expenditure, in euros
R&D effort (RDe)	R&D expenditure as a percentage of sales
Subcontracting of research activities (subcontract_R)	Percentage of research activities subcontracted by the firm
Subcontracting of technological activities (subcontract_D)	Percentage of development activities subcontracted by the firm
Firms submitting several projects simultaneously (multiproject)	Dummy variable indicating if the firm submits several R&D projects to the agency
Firms applying for R&D subsidies in 2005 and 2006 (repeat)	Dummy variable indicating if the firm applies for subsidies in the two years of the sample
Project-level	
Proposed cost of the project (cost)	Expected cost of the project according to the firm, in euros
Subsidised project (receive)	Dummy variable indicating whether the project has been selected to be subsidised
Project score for technological contribution (tech_contrib)	Technological contribution of the project, values from 0 to 175
Project score for socio-economic impact (se_impact)	Socio-economic impact of the project, values from 0 to 125
Project score for internationalisation (international)	Contribution of the project to the internationalisation of the firm, values from 0 to 75
Project score for diffusion (diff)	Impact of the project on other sectors and/or firms, values from 0 to 75
Overall project score (total)	Sum of previous scores, values from 0 to 450
Subsidy-level	
Awardable cost of the project (awardable_cost)	Expected cost of the project according to the agency, in euros
Amount of the subsidy (subsidy)	Subsidy conceded to the project, in euros
Subsidy's intensity (RDi)	R&D subsidy as a percentage of awardable cost of the project

	Total (968 firms)				Accepted (551 firms)				Rejected (417 firms)						
	Mean	Median	Std. Dev.	Min	Max	Mean	Median	Std. Dev.	Min	Max	Mean	Median	Std. Dev.	Min	Max
emp	196.5	54.0	524.5	1	6697	250.5	64.0	645.9	2	6697	125.1	47.0	281.1	1	3009
age	26.4	21.0	22.2	1	171	28.6	23.0	23.4	1	171	23.4	20.0	20.2	1	139
dloc	0.2	0.0	0.4	0	1	0.2	0.0	0.4	0	1	0.2	0.0	0.4	0	1
dht	0.1	0.0	0.3	0	1	0.1	0.0	0.3	0	1	0.1	0.0	0.3	0	1
sht	0.1	0.0	0.3	0	1	0.1	0.0	0.3	0	1	0.2	0.0	0.4	0	1
<i>exports</i> _{t-1}	0.3	0.2	0.3	0	1	0.3	0.3	0.3	0	1	0.2	0.1	0.3	0	1
$Patents_{t-1}$	13.6	0.0	138.1	0	3000	20.2	0.0	180.5	0	3000	4.7	0.0	33.6	0	536
RDdept	0.8	1.0	0.4	0	1	0.8	1.0	0.4	0	1	0.7	1.0	0.4	0	1
RDemployment _{t-1}	0.1	0.1	0.2	0	1	0.1	0.1	0.2	0	1	0.1	0.1	0.1	0	1
<i>RDexpenditure</i> _{t-1}	1067.5	250.0	4190.0	0	80678	1342.4	304.0	4950.1	0	80678	696.0	170.1	2822.6	0	48413
RDe	0.1	0.0	0.2	0	0.9	0.1	0.0	0.2	0	0.9	0.1	0.0	0.2	0	0.9
subcontract_ R_{t-1}	0.1	0.0	0.2	0	1	0.1	0.1	0.2	0	1	0.1	0.0	0.2	0	1
$subcontract_D_{t-1}$	0.1	0.1	0.2	0	1	0.1	0.1	0.2	0	1	0.1	0.1	0.2	0	1
multiproject	0.2	0.0	0.4	0	1	0.2	0.0	0.4	0	1	0.1	0.0	0.3	0	1
repeat	0.5	0.0	0.5	0	1	0.6	1.0	0.5	0	1	0.3	0.0	0.5	0	1
cost	989.4	342.1	3763.3	21.6	78664.2	1321.7	393.0	4907.4	60.4	78664.2	550.4	301.6	860.9	21.6	10245.6
tech_contrib	79.8	77.0	36.3	0	168	95.0	91.0	31.8	0.0	168	59.7	62.0	31.8	0	154
se_Impact	36.4	32.3	24.1	0	125	47.2	40.1	24.4	0.0	125	22.0	21.1	14.1	0	79
international	23.2	13.5	21.8	0	75	31.2	27.0	23.5	0.0	75	12.8	8.0	13.4	0	75
diff	17.7	13.0	15.0	0	75	22.3	21.0	15.1	0.0	75	11.6	6.0	12.5	0	62
total	234.7	255.0	82.7	0	422.6	288.9	278.0	38.4	0.0	422.6	163.1	173.0	70.2	0	389.7
awardable_cost	143.1	73.1	273.7	0	3744.1	251.4	170.0	323.1	60	3744.1	0.0	0.0	0.0	0	0
subsidy	32.3	16.2	55.4	0	645.9	56.7	38.4	63.3	6.1	645.9	0.0	0.0	0.0	0	0
RDi	0.1	0.2	0.1	0	0.6	0.2	0.2	0.1	0.1	0.6	0.0	0.0	0.0	0	0

Table 2. Descriptive statistics

Variables	No	Total project	Detailed project scores		
	scores	scores			
1()	-0.0038	-0.0083	-0.0113		
log(emp)	(-0.66)	(-2.40)**	(-2.62)***		
1()	0.0184	0.0113	0.0201		
log(age)	(1.85)*	(1.92)*	(2.72)***		
11	-0.0284	-0.0166	-0.0137		
aloc	(-1.46)	(-1.41)	(-0.94)		
alt	0.0363	-0.0050	0.0209		
Sht	(1.46)	(-0.33)	(1.11)		
-11-4	-0.0102	0.0178	0.0043		
ani	(-0.41)	(1.21)	(0.23)		
	0.0454	-0.0122	0.0055		
$exports_{t-1}$	(1.80)*	(-0.83)	(0.30)		
log(natouta)	0.0067	0.0042	-0.0057		
$log(patents)_{t-1}$	(1.07)	(1.16)	(-1.21)		
DDdomt	0.0334	-0.0102	0.0020		
кДаері	(1.74)*	(-0.86)	(0.34)		
auto contugat D	0.0577	0.0117	0.0615		
SUDCONIFACI_K _{t-1}	(1.43)	(0.47)	(2.02)**		
nonost	0.0868	0.0225	0.428		
repeat	(5.87)***	(2.55)**	(3.90)***		
total		0.0024			
lolal	,	(28.50)***			
took oontuik			0.0028		
lecn_contrib			(15.04)***		
sa impact			0.0011		
se_impaci			(4.49)***		
intomation al			0.0027		
international			(7.73)***		
1:4			0.0019		
aijj	,		(3.85)***		
constant	-0.0568	-0.5258	-0.3426		
considiti	(-1.72)*	(-17.42)***	(-11.13)		
N	968	968	968		
R^2	0.128	0.557	0.438		
$LR-\chi^2$	68.65***	1040.69***	623.36***		

Table 3. First stage Tobit results. Dependent variable: percentage of public subsidy received (%RDi).

Notes: *t*-values in parenthesis. *, ** and *** indicate significance at 10, 5 and 1 percent levels, respectively.

Variables	Determinants of receiving a public subsidy or not				
log(own)	-0.0295				
log(emp)	(-0.66)				
log(age)	0.0874				
log(age)	(1.42)				
log (PD own on dituno)	0.1193				
log(KDexpenditure) _{t-1}	(3.04)***				
dlaa	-0.2482				
	(-2.13)**				
abt	0.0808				
Sni	(0.54)				
11.4	-0.1915				
ant	(-1.24)				
	0.2244				
exports _{t-1}	(1.43)				
1(0.0309				
log(patents) _{t-1}	(0.74)				
	0.0267				
RDaept	(0.22)				
	0.5139				
repeat	(5.77)***				
	-0.8982				
constant	(-4.15)***				
Ν	931				
$Pseudo-R^2$	0.0719				
$LR-\gamma^{2}(11)$	91.32***				

Table 4. The selection equation. The first stage of the Heckman procedure (probit model). Endogenous variable *receive*.

Notes: *t*-values in parenthesis and *p*-values in brackets. *, ** and *** indicate significance at 10, 5 and 1 percent levels, respectively.

Variables	0	LS	Heckman 2-stage estimator			
mucd(0/DDi)	0.0405	0.0362	0.0582	0.0848		
prea(%RDI)	(0.98)	(0.25)	(1.41)	(0.60)		
$pund(0/RDi)^2$		0.0098		-0.0600		
pred(%KDI)		(0.03)		(-0.20)		
log(amp)	-0.0476	-0.0476	-0.0489	-0.0489		
log(emp)	(-11.24)***	(-11.20)***	(-10.87)***	(-10.83)***		
100(000)	-0.0270	-0.0270	-0.0233	-0.0234		
log(age)	(-4.95)***	(4.92)***	(-3.86)***	(-3.86)***		
log (PD own on dituma)	0.0315	0.0315	0.0373	0.0373		
log(KDexpenditure) _{t-1}	(7.98)***	(7.97)***	(7.73)***	(7.72)***		
dlag	0.0306	0.0306	0.0211	0.0210		
aloc	(2.79)***	(2.79)***	(1.72)*	(1.70)*		
abt	0.0430	0.0431	0.0424	0.0423		
Sni	(3.06)***	(3.06)***	(2.85)***	(2.84)***		
dlat	-0.0027	-0.0027	-0.0092	-0.0093		
ani	(-0.21)	(-0.21)	(-0.63)	(-0.64)		
ownowta	-0.0227	-0.0227	-0.0111	-0.0111		
$expons_{t-1}$	(-1.77)*	(-1.77)*	(-0.76)	(-0.76)		
log(ngtouts)	0.0035	0.0035	0.0041	0.0041		
log(palents) _{t-1}	(1.08)	(1.08)	(1.17)	(1.17)		
PDdant	0.0030	0.0031	0.0072	0.0070		
краері	(0.22)	(0.23)	(0.51)	(0.50)		
PDamployment	0.0892	0.0892	0.0896	0.0895		
KDemployment _{t-1}	(3.56)***	(3.55)***	(3.62)***	(3.62)***		
subcontract P	0.0074	0.0074	0.0122	0.0120		
subcontract_R _{t-1}	(0.32)	(0.32)	(0.54)	(0.53)		
subcontract D.	-0.0085	-0.0085	-0.0057	-0.0056		
subcommuci_D _{t-1}	(-0.41)	(-0.41)	(-0.28)	(-0.28)		
multiproject	0.0015	0.0015	0.0025	0.0026		
muniprojeci	(0.16)	(0.16)	(0.26)	(0.27)		
constant	0.1585	0.1588	-0.8982	-0.8982		
constant	(6.79)***	(6.22)***	(4.15)***	(4.15)***		
Mills ())			0.0668	0.0673		
Millis (N)			(2.33)***	(2.34)**		
N	535	535	931	931		
Censored obs.	,		396	396		
Uncensored obs.	,		535	535		
R^2	0.497	0.497				
$LR-\chi^2$			432.88***	432.06***		

Table 5. The impact of subsidy intensity (%*RDi*) on firms' R&D effort (ln*RDe*)

Notes: for OLS estimation *t*-ratio in parenthesis, for Heckman 2-stage estimation *z*-values in parenthesis. *, ** and *** indicate significance at 10, 5 and 1 percent levels, respectively. Predicted values obtained from the Tobit model with the detailed project scores (column 3 in Table 3).





Figure 2. Non-parametric conditional quantile regression controlling for possible problems of endogeneity.





SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

2006

CREAP2006-01 Matas, A. (GEAP); Raymond, J.Ll. (GEAP) "Economic development and changes in car ownership patterns" (Juny 2006)

CREAP2006-02

Trillas, F. (IEB); **Montolio, D.** (IEB); **Duch, N.** (IEB) "Productive efficiency and regulatory reform: The case of Vehicle Inspection Services" (Setembre 2006)

CREAP2006-03

Bel, G. (PPRE-IREA); **Fageda, X.** (PPRE-IREA) "Factors explaining local privatization: A meta-regression analysis" (Octubre 2006)

CREAP2006-04

Fernàndez-Villadangos, L. (PPRE-IREA) "Are two-part tariffs efficient when consumers plan ahead?: An empirical study" (Octubre 2006)

CREAP2006-05

Artís, M. (AQR-IREA); Ramos, R. (AQR-IREA); Suriñach, J. (AQR-IREA) "Job losses, outsourcing and relocation: Empirical evidence using microdata" (Octubre 2006)

CREAP2006-06

Alcañiz, M. (RISC-IREA); Costa, A.; Guillén, M. (RISC-IREA); Luna, C.; Rovira, C. "Calculation of the variance in surveys of the economic climate" (Novembre 2006)

CREAP2006-07

Albalate, D. (PPRE-IREA) "Lowering blood alcohol content levels to save lives: The European Experience" (Desembre 2006)

CREAP2006-08 Garrido, A. (IEB); Arqué, P. (IEB)

"The choice of banking firm: Are the interest rate a significant criteria?" (Desembre 2006)



SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

CREAP2006-09 Segarra, A. (GRIT); **Teruel-Carrizosa, M.** (GRIT) "Productivity growth and competition in spanish manufacturing firms: What has happened in recent years?" (Desembre 2006)

CREAP2006-10 Andonova, V.; Díaz-Serrano, Luis. (CREB) "Political institutions and the development of telecommunications" (Desembre 2006)

CREAP2006-11 Raymond, J.L.(GEAP); **Roig, J.L.** (GEAP) "Capital humano: un análisis comparativo Catalunya-España" (Desembre 2006)

CREAP2006-12 Rodríguez, M.(CREB); **Stoyanova, A.** (CREB) "Changes in the demand for private medical insurance following a shift in tax incentives" (Desembre 2006)

CREAP2006-13 Royuela, V. (AQR-IREA); **Lambiri, D.**; **Biagi**, **B.** "Economía urbana y calidad de vida. Una revisión del estado del conocimiento en España" (Desembre 2006)

CREAP2006-14 Camarero, M.; Carrion-i-Silvestre, J.LL. (AQR-IREA).;Tamarit, C. "New evidence of the real interest rate parity for OECD countries using panel unit root tests with breaks" (Desembre 2006)

CREAP2006-15 Karanassou, M.; Sala, H. (GEAP).;Snower, D. J. "The macroeconomics of the labor market: Three fundamental views" (Desembre 2006)



2007

XREAP2007-01

Castany, L (AQR-IREA); **López-Bazo, E.** (AQR-IREA).;**Moreno**, **R.** (AQR-IREA) "Decomposing differences in total factor productivity across firm size" (Març 2007)

XREAP2007-02

Raymond, J. Ll. (GEAP); **Roig, J. Ll.** (GEAP) "Una propuesta de evaluación de las externalidades de capital humano en la empresa" (Abril 2007)

XREAP2007-03

Durán, J. M. (IEB); **Esteller, A.** (IEB) "An empirical analysis of wealth taxation: Equity vs. Tax compliance" (Juny 2007)

XREAP2007-04

Matas, A. (GEAP); Raymond, J.Ll. (GEAP) "Cross-section data, disequilibrium situations and estimated coefficients: evidence from car ownership demand" (Juny 2007)

XREAP2007-05

Jofre-Montseny, J. (IEB); **Solé-Ollé, A.** (IEB) "Tax differentials and agglomeration economies in intraregional firm location" (Juny 2007)

XREAP2007-06

Álvarez-Albelo, C. (CREB); Hernández-Martín, R.

"Explaining high economic growth in small tourism countries with a dynamic general equilibrium model" (Juliol 2007)

XREAP2007-07

Duch, N. (IEB); Montolio, D. (IEB); Mediavilla, M.

"Evaluating the impact of public subsidies on a firm's performance: a quasi-experimental approach" (Juliol 2007)

XREAP2007-08

Segarra-Blasco, A. (GRIT) "Innovation sources and productivity: a quantile regression analysis" (Octubre 2007)



SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

XREAP2007-09 Albalate, D. (PPRE-IREA) "Shifting death to their Alternatives: The case of Toll Motorways" (Octubre 2007)

XREAP2007-10 Segarra-Blasco, A. (GRIT); Garcia-Quevedo, J. (IEB); Teruel-Carrizosa, M. (GRIT) "Barriers to innovation and public policy in catalonia" (Novembre 2007)

XREAP2007-11
Bel, G. (PPRE-IREA); Foote, J.
"Comparison of recent toll road concession transactions in the United States and France" (Novembre 2007)

XREAP2007-12 Segarra-Blasco, A. (GRIT); "Innovation, R&D spillovers and productivity: the role of knowledge-intensive services" (Novembre 2007)

XREAP2007-13 Bermúdez Morata, Ll. (RFA-IREA); Guillén Estany, M. (RFA-IREA), Solé Auró, A. (RFA-IREA) "Impacto de la inmigración sobre la esperanza de vida en salud y en discapacidad de la población española" (Novembre 2007)

XREAP2007-14 Calaeys, P. (AQR-IREA); Ramos, R. (AQR-IREA), Suriñach, J. (AQR-IREA) "Fiscal sustainability across government tiers" (Desembre 2007)

XREAP2007-15 Sánchez Hugalbe, A. (IEB) "Influencia de la inmigración en la elección escolar" (Desembre 2007)



2008

XREAP2008-01

Durán Weitkamp, C. (GRIT); **Martín Bofarull, M.** (GRIT) ; **Pablo Martí, F.** "Economic effects of road accessibility in the Pyrenees: User perspective" (Gener 2008)

XREAP2008-02

Díaz-Serrano, L.; Stoyanova, A. P. (CREB)

"The Causal Relationship between Individual's Choice Behavior and Self-Reported Satisfaction: the Case of Residential Mobility in the EU" (Març 2008)

XREAP2008-03

Matas, A. (GEAP); Raymond, J. L. (GEAP); Roig, J. L. (GEAP) "Car ownership and access to jobs in Spain" (Abril 2008)

XREAP2008-04

Bel, G. (PPRE-IREA) **; Fageda, X.** (PPRE-IREA) "Privatization and competition in the delivery of local services: An empirical examination of the dual market hypothesis" (Abril 2008)

XREAP2008-05

Matas, A. (GEAP); **Raymond, J. L.** (GEAP); **Roig, J. L.** (GEAP) "**Job accessibility and employment probability**" (Maig 2008)

XREAP2008-06

Basher, S. A.; Carrión, J. Ll. (AQR-IREA) Deconstructing Shocks and Persistence in OECD Real Exchange Rates (Juny 2008)

XREAP2008-07

Sanromá, E. (IEB); **Ramos, R.** (AQR-IREA); Simón, H. Portabilidad del capital humano y asimilación de los inmigrantes. Evidencia para España (Juliol 2008)

XREAP2008-08 Basher, S. A.; Carrión, J. Ll. (AQR-IREA) Price level convergence, purchasing power parity and multiple structural breaks: An application to US cities

(Juliol 2008)

XREAP2008-09 Bermúdez, Ll. (RFA-IREA) A priori ratemaking using bivariate poisson regression models (Juliol 2008)



SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

XREAP2008-10 Solé-Ollé, A. (IEB), Hortas Rico, M. (IEB) Does urban sprawl increase the costs of providing local public services? Evidence from Spanish municipalities (Novembre 2008)

XREAP2008-11 Teruel-Carrizosa, M. (GRIT), Segarra-Blasco, A. (GRIT) Immigration and Firm Growth: Evidence from Spanish cities (Novembre 2008)

XREAP2008-12

Duch-Brown, N. (IEB), **García-Quevedo, J.** (IEB), **Montolio, D.** (IEB) Assessing the assignation of public subsidies: Do the experts choose the most efficient R&D projects? (Novembre 2008)

XREAP2008-13 Bilotkach, V., Fageda, X. (PPRE-IREA), Flores-Fillol, R. Scheduled service versus personal transportation: the role of distance (Desembre 2008)

XREAP2008-14 Albalate, D. (PPRE-IREA), Gel, G. (PPRE-IREA) Tourism and urban transport: Holding demand pressure under supply constraints (Desembre 2008)



2009

XREAP2009-01 Calonge, S. (CREB); Tejada, O. "A theoretical and practical study on linear reforms of dual taxes"

XREAP2009-02

(Febrer 2009)

Albalate, D. (PPRE-IREA); Fernández-Villadangos, L. (PPRE-IREA) "Exploring Determinants of Urban Motorcycle Accident Severity: The Case of Barcelona" (Març 2009)

XREAP2009-03

Borrell, J. R. (PPRE-IREA); **Fernández-Villadangos, L.** (PPRE-IREA) "Assessing excess profits from different entry regulations" (Abril 2009)

XREAP2009-04 Sanromá, E. (IEB); Ramos, R. (AQR-IREA), Simon, H. "Los salarios de los inmigrantes en el mercado de trabajo español. ¿Importa el origen del capital humano?" (Abril 2009)

XREAP2009-05 Jiménez, J. L.; Perdiguero, J. (PPRE-IREA) "(No)competition in the Spanish retailing gasoline market: a variance filter approach" (Maig 2009)

XREAP2009-06 Álvarez-Albelo, C. D. (CREB), Manresa, A. (CREB), Pigem-Vigo, M. (CREB) "International trade as the sole engine of growth for an economy" (Juny 2009)

XREAP2009-07 Callejón, M. (PPRE-IREA), Ortún V, M. "The Black Box of Business Dynamics" (Setembre 2009)

XREAP2009-08 Lucena, A. (CREB) "The antecedents and innovation consequences of organizational search: empirical evidence for Spain" (Octubre 2009)

XREAP2009-09 Domènech Campmajó, L. (PPRE-IREA) "Competition between TV Platforms" (Octubre 2009)



SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

XREAP2009-10

Solé-Auró, A. (RFA-IREA),**Guillén, M.** (RFA-IREA), **Crimmins, E. M.** "Health care utilization among immigrants and native-born populations in 11 European countries. Results from the Survey of Health, Ageing and Retirement in Europe" (Octubre 2009)

XREAP2009-11 Segarra, A. (GRIT), Teruel, M. (GRIT) "Small firms, growth and financial constraints" (Octubre 2009)

XREAP2009-12

Matas, A. (GEAP), **Raymond, J.Ll.** (GEAP), **Ruiz, A.** (GEAP) "Traffic forecasts under uncertainty and capacity constraints" (Novembre 2009)

XREAP2009-13

Sole-Ollé, A. (IEB) "Inter-regional redistribution through infrastructure investment: tactical or programmatic?" (Novembre 2009)

XREAP2009-14 Del Barrio-Castro, T., García-Quevedo, J. (IEB) "The determinants of university patenting: Do incentives matter?" (Novembre 2009)

XREAP2009-15

Ramos, R. (AQR-IREA), **Suriñach, J.** (AQR-IREA), **Artís, M.** (AQR-IREA) "Human capital spillovers, productivity and regional convergence in Spain" (Novembre 2009)

XREAP2009-16 Álvarez-Albelo, C. D. (CREB), Hernández-Martín, R. "The commons and anti-commons problems in the tourism economy" (Desembre 2009)



2010

XREAP2010-01

García-López, M. A. (GEAP) "The Accessibility City. When Transport Infrastructure Matters in Urban Spatial Structure" (Febrer 2010)

XREAP2010-02

García-Quevedo, J. (IEB), **Mas-Verdú, F.** (IEB), **Polo-Otero, J.** (IEB) "Which firms want PhDs? The effect of the university-industry relationship on the PhD labour market" (Març 2010)

XREAP2010-03

Pitt, D., Guillén, M. (RFA-IREA) "An introduction to parametric and non-parametric models for bivariate positive insurance claim severity distributions" (Març 2010)

XREAP2010-04

Bermúdez, Ll. (RFA-IREA), **Karlis, D.** "Modelling dependence in a ratemaking procedure with multivariate Poisson regression models" (Abril 2010)

XREAP2010-05

Di Paolo, A. (IEB) "Parental education and family characteristics: educational opportunities across cohorts in Italy and Spain" (Maig 2010)

XREAP2010-06

Simón, H. (IEB), Ramos, R. (AQR-IREA), Sanromá, E. (IEB) "Movilidad ocupacional de los inmigrantes en una economía de bajas cualificaciones. El caso de España" (Juny 2010)

XREAP2010-07

Di Paolo, A. (GEAP & IEB), **Raymond, J. Ll.** (GEAP & IEB) "Language knowledge and earnings in Catalonia" (Juliol 2010)

XREAP2010-08

Bolancé, C. (RFA-IREA), **Alemany, R.** (RFA-IREA), **Guillén, M.** (RFA-IREA) "Prediction of the economic cost of individual long-term care in the Spanish population" (Setembre 2010)

XREAP2010-09

Di Paolo, A. (GEAP & IEB) "Knowledge of catalan, public/private sector choice and earnings: Evidence from a double sample selection model" (Setembre 2010)



SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

XREAP2010-10

Coad, A., Segarra, A. (GRIT), **Teruel, M.** (GRIT) "Like milk or wine: Does firm performance improve with age?" (Setembre 2010)

XREAP2010-11 Di Paolo, A. (GEAP & IEB), Raymond, J. Ll. (GEAP & IEB), Calero, J. (IEB) "Exploring educational mobility in Europe" (Octubre 2010)

XREAP2010-12

Borrell, A. (GiM-IREA), **Fernández-Villadangos, L.** (GiM-IREA) "Clustering or scattering: the underlying reason for regulating distance among retail outlets"

(Desembre 2010)

XREAP2010-13

Di Paolo, A. (GEAP & IEB) "School composition effects in Spain" (Desembre 2010)

XREAP2010-14

Fageda, X. (GiM-IREA), **Flores-Fillol, R.** "Technology, Business Models and Network Structure in the Airline Industry" (Desembre 2010)

XREAP2010-15

Albalate, D. (GiM-IREA), Bel, G. (GiM-IREA), Fageda, X. (GiM-IREA) "Is it Redistribution or Centralization? On the Determinants of Government Investment in Infrastructure" (Desembre 2010)

XREAP2010-16Oppedisano, V., Turati, G."What are the causes of educational inequalities and of their evolution over time in Europe? Evidence from PISA"(Desembre 2010)

XREAP2010-17

Canova, L., Vaglio, A. "Why do educated mothers matter? A model of parental help" (Desembre 2010)



2011

XREAP2011-01

Fageda, X. (GiM-IREA), **Perdiguero, J.** (GiM-IREA) "An empirical analysis of a merger between a network and low-cost airlines" (Maig 2011)

XREAP2011-02

Moreno-Torres, I. (ACCO, CRES & GiM-IREA) "What if there was a stronger pharmaceutical price competition in Spain? When regulation has a similar effect to collusion" (Maig 2011)

XREAP2011-03

Miguélez, E. (AQR-IREA); **Gómez-Miguélez, I.** "Singling out individual inventors from patent data" (Maig 2011)

XREAP2011-04 Moreno-Torres, I. (ACCO, CRES & GiM-IREA) "Generic drugs in Spain: price competition vs. moral hazard" (Maig 2011)

XREAP2011-05

Nieto, S. (AQR-IREA), Ramos, R. (AQR-IREA) "¿Afecta la sobreeducación de los padres al rendimiento académico de sus hijos?" (Maig 2011)

XREAP2011-06

Pitt, D., Guillén, M. (RFA-IREA), **Bolancé, C.** (RFA-IREA) "Estimation of Parametric and Nonparametric Models for Univariate Claim Severity Distributions - an approach using R" (Juny 2011)

XREAP2011-07

Guillén, M. (RFA-IREA), **Comas-Herrera, A.** "How much risk is mitigated by LTC Insurance? A case study of the public system in Spain" (Juny 2011)

XREAP2011-08 Ayuso, M. (RFA-IREA), Guillén, M. (RFA-IREA), Bolancé, C. (RFA-IREA) "Loss risk through fraud in car insurance" (Juny 2011)

XREAP2011-09 Duch-Brown, N. (IEB), García-Quevedo, J. (IEB), Montolio, D. (IEB) "The link between public support and private R&D effort: What is the optimal subsidy?" (Juny 2011)



xreap@pcb.ub.es